

∂ Open access • Journal Article • DOI:10.1080/02692171.2015.1054366

Financial openness and financial development: an analysis using indices — Source link

Zeynep Ozkok

Institutions: St. Francis Xavier University

Published on: 22 Jun 2015 - International Review of Applied Economics (Routledge)

Topics: Financial ratio, Financial analysis, Financial modeling, Market data and Indirect finance

Related papers:

- Openness and Financial Development in China: The Political Economy of Financial Resources Distribution
- Financial Development, Openness and Institutions: Evidence from Panel Data
- Does Openness Matter for Financial Development in Africa
- Trends in Financial Integration and Financial Development in Selected Asian Countries: Is There Any Relationship?
- The determinants of financial integration





Financial Openness and Financial Development: An Analysis Using Indices

Ozkok, Zeynep

St. Francis Xavier University

10 December 2010

Online at https://mpra.ub.uni-muenchen.de/58865/ MPRA Paper No. 58865, posted 25 Sep 2014 23:45 UTC

Financial Openness and Financial Development:

An Analysis Using Indices

Zeynep Ozkok*

St. Francis Xavier University

(2010)

Abstract

This paper examines the link between financial openness and financial development through panel data analysis on advanced and emerging market countries. Using indices, financial openness together with institutional and educational variables explains a large part of the variation in financial development across countries and over time. Our analysis demonstrates that different indexing strategies serve in finding better measures for financial openness and financial development in comparison to the individual indicators used in the literature. Our principal component type financial openness index conveys a positive effect on financial development independent from the lag structure or specifications used.

JEL Codes: F36, F63, F65, O16

Keywords: Financial liberalization; Financial development; Financial openness; Financial markets; Principal component analysis; Dynamic panel data analysis

^{*} Corresponding author address: Department of Economics PO Box 5000, Antigonish, NS, B2G 2W5, Canada. Email: <u>zozkok@stfx.ca</u>. Telephone: + 1 (902) 867 5855

1. Introduction

In the wake of the recent financial crisis, the role of financial development in emerging markets and developed countries has gained further interest among researchers. Financial development, the channel for increasing the efficiency of financial markets and resources and improving on the overall importance of the financial system, has become one of the main elements that influences economic growth and welfare (Huang, 2006).

Although the role of financial development on economic growth is newly recognized, a wide spread debate on the effects of financial liberalization on growth and financial development is ongoing. Studies have shown that financial liberalization through alleviation of capital controls and deregulation of financial markets can endorse economic growth and enhance welfare by creating opportunities for a better and more efficient allocation of resources and by portfolio and risk diversification under appropriate controls, frameworks and regulatory apparatus (Aziakpono, 2007; Chinn and Ito, 2006).

The literature provides a broad examination of financial liberalization and economic growth, nevertheless, the link between financial liberalization and financial development has been overlooked. A proper analysis of this link will help clarify the ambiguity in the relationship between financial liberalization and economic growth. Our paper, by this means, examines the effects of financial openness, the most prominent measure of financial liberalization, and financial development through a panel study of developed and emerging market countries. Using indices we show that financial openness together with institutional and educational variables explains a large part of the variation in financial development across countries and over time. Principal component type indices provide better results in terms of economic and statistical significances. The robustness checks and the heterogeneity analysis convey that findings are robust to different lag structures, specifications and reductions in the sample size.

The small strand of literature that we base our analysis on examines the effects of trade and capital account openness on financial development. Chinn and Ito (2002) and (2006), Ito (2006), Baltagi, Demetriades and Law (2007), Demetriades and Law (2006), Demetriades and Andrianova (2005) and Huang (2006) show that financial liberalization (capital account openness in most cases) contributes to financial development in equity and stock markets for both less developed and emerging market countries. There are, however, three main issues in examining the link between financial openness and financial development. First, the choice of indicators has been a topic of concern. Studies lack a comprehensive indicator that can bring together all features of financial development; the banking system, the stock and the bond markets. With different measures used for financial openness and for financial development, the results obtained seem unconvincing. Another concern is that the results from various studies become hardly comparable due to the particular choice of individual measures and due to the country and time coverage selected for the study. Constructing better financial openness and financial development indices will help resolve problems associated with particular choice of measures. Second, the number of countries included in most studies is limited. Due to the lack of data for many less developed and some emerging market countries, most studies use developed countries in their estimations, which highly influences the results. Lastly, what seems to be a minor issue, which in reality can affect almost all findings, is the choice of control variables. The literature shows that the choice of control variables can influence the link between financial openness and financial development. Correct specification of control variables can lead to a better examination of these concepts.

Compared to the existing literature we contribute in three aspects. First, we give a view on different indices for financial openness and financial development straining away from choosing individual variables which we believe do not fully represent the aspects of financial openness and financial development. Second, we examine the simultaneity hypothesis of opening financial and goods markets with indices. Lastly we explicitly study one of the main problems of panel data models; heterogeneity issues.

This paper is organized as follows. The next section introduces the data. Section 3 describes the construction of aggregate indices of financial openness and financial development. Section 4 provides a discussion of the empirical model. Section 5 reports our estimation results. Section 6 discusses the robustness checks and further issues related to our sample. Section 7 concludes.

2. Data

The analysis is based on annual data for 61 developing and advanced countries, over a 12 year period of 1996 - 2007. The data are obtained primarily from Beck, Demirguc-Kunt, and Levine's database on *Financial Development and Structure* (referred to as *BDL* from onwards), the World Bank's *World Development Indicators (WDI)*, *Worldwide Governance Indicators*, and *Edstats* which extracts data from the UNESCO Institute for Statistics.

2.1 Financial Openness Indicators

Financial openness is measured with *foreign direct investment (% of GDP)*, *portfolio investment flows (% of GDP)*, and *international debt issues (% of GDP)*.

Foreign direct investment, is the sum of net inflows and outflows of foreign direct investment recorded as a percentage of GDP. This indicator adds up equity capital, reinvestment of earnings and other short- and long-term capital (World Bank, 2007). *Portfolio investment flows (% of GDP)*, is the sum of portfolio debt flows (private and publicly guaranteed and private nonguaranteed bond issues purchased by foreign investors) and non-debt-creating portfolio equity flows. ¹ *International debt issues (% of GDP)* measures "the net flow of international bond issues relative to a country's economic activity." (Beck and Demirguc-Kunt, 2009, 15).²

As summarized by Kose et al. (2006) financial openness indicators are divided into two mainstream measures; de jure measures which depend on the removal of legal restrictions and controls on cross-border capital flows, prices, quantities and foreign equity holdings; and de facto measures which observe countries' actual integration into the world capital and financial markets through flow variables. De jure measures such as the IMF's Annual Report on Exchange Arrangements and Exchange Restrictions (AREAER) typically illustrate the number of years for which a country's capital accounts have been open and free from restrictions and controls. The AREAER measure, Chinn and Ito's (2005) capital openness measure, Quinn's capital account openness index (1997, 2003), Mody and Murshid's (2005) and Edwards' (2004) measures on capital and current account restrictions are based on narrative, discrete 0-1 type variables that demonstrate openness or closedness of capital accounts (Bussiere and Fratzscher, 2004). This type of measures have long been accused of not fully reflecting the degree of financial or capital account openness due to their reliance on the removal of restrictions associated with foreign exchange transactions (Kose et al. 2006). Although they aim to measure financial globalization in terms of openness of capital and financial markets, they do not represent the extent of integration into the global markets. Alternatively, de facto measures focus on legal restrictions and capital flows. Although these measures may cause measurement errors and create difficulty in overcoming endogenity and causality issues, they remain to be the superior measure of financial integration (Kose et al., 2006). For all these reasons, we restrain from using discrete de jure type measures and prefer to use stock and flow variables to measure financial openness.³

2.2 Financial Development Indicators

Financial development indicators consist of banking system, stock market and bond market measures.

Banking sector development indicators:

Liquid liabilities (% of GDP) equals the ratio of liquid liabilities of bank and nonbank financial intermediaries to GDP (Demirguc-Kunt, Levine, 2001). This variable is commonly used as a measure of financial sector development and is a typical measure of financial depth.

¹ Non-debt-creating portfolio equity flows are defined as the sum of country funds, depository receipts, and direct purchases of shares by foreign investors (World Bank, 2007).

² The literature suggests the use of gross foreign direct investment, gross private capital flows, and some de jure type indices as measures of financial openness. Gross private capital flows are excluded from our analysis and are replaced by portfolio investment flows due to their discontinuity by the World Bank.

³ We only report the results for one of the most frequently used de jure type indices, the Chinn and Ito index of capital openness, for comparison reasons. We believe that these results guide in demonstrating the strength of our de facto type financial openness index in comparison to the individual de facto measures by the literature.

Private credit by deposit money banks and other institutions (% of GDP) is an indicator for the overall development in private banking markets (Chinn and Ito, 2006). This variable includes financial resources provided to the private sector by deposit money banks and other financial institutions. It measures the credit provided to the private sector.

The ratio of deposit money bank assets to the sum of deposit money bank assets and central bank assets (in *percentages*) is used to demonstrate the weight of deposit money bank assets among total assets. It reflects the importance of private lending compared to total lending (Huang, 2006).

Total bank assets (% of GDP) is used as a measure of financial depth. It represents the overall size of the banking sector.

Domestic credit provided by the banking sector (% of GDP) includes credit extended to the private sector and general government, to the nonfinancial public sector in the form of investments in short- and long-term government securities, to banking and nonbank institutions and includes loans to state enterprises but excludes credit to the central government (World Bank, 2007). It is a measure of banking sector depth and financial sector development in terms of size (World Bank, 2007).

The variables for banking sector development correspond to those used in the literature. We believe that a wide range of different variables will help capture all aspects of banking sector development.

Stock market development indicators:

Stock market capitalization (% of GDP) is equal to the value of listed shares divided by GDP. It is an indicator of the size of the stock market. Stock market turnover ratio (in percentages) is used as the efficiency indicator of stock markets (Demirguc-Kunt and Levine, 2001). It is classified as the ratio of the value of total shares traded to stock market capitalization. Stock market total value traded (% of GDP) is equal to the total shares traded on the stock market exchange divided by GDP. This indicator measures the activity or liquidity of the stock markets (Demirguc-Kunt and Levine, 2001).

Stock market development indicators used correspond to the ones from the literature. We believe that these variables will summarize all prospects of stock market development.

Bond market development indicators:

Private bond market capitalization (% of GDP) and *public bond market capitalization (% of GDP)* are the two indicators used to measure bond market development.

Private bond market capitalization (% of GDP) is equal to the total amount of outstanding domestic debt securities issued by financial institutions and corporations as a share of GDP. *Public bond market capitalization (% of GDP)* is equal to the total amount of public domestic securities issued by governments as a share of GDP. Both of these indicators are used to determine the efficiency of bond markets.

Bond market development indicators have not been previously used in the literature on financial openness and financial development. Even though they have been employed extensively in the equity market development literature, due to their short period of availability they have not been used as indicators for financial development. In our analysis we propose using the bond market development indicators in order to capture the efficiency and the effectiveness of bond markets on the overall level of financial development.

2.3 Control Variables

To examine the effect of financial openness on financial development we introduce a broad range of control variables. The control variables include *GDP* per capita, trade openness, secondary school enrollment rate, and legal and institutional variables. *Logarithm of GDP per capita (in constant 2000 US dollars)* is used as a measure of economic performance among countries. We employ this measure to control for the demand of finance and to monitor the differences in performances and productivities across countries.⁴

Trade openness (% of GDP) measured by the sum of imports and exports of goods and services is used to determine whether trade liberalization is a precondition for financial liberalization. Controlling for trade openness allows examining the direct effects of financial liberalization on financial development.

Secondary school gross enrollment rate (% of population) is used as an indicator that controls for differences in educational attainment across countries. We consider this measure as an important reason for why we observe disparities across countries in their levels of financial development. Even though an educational attainment indicator has not been used previously as a control variable in the financial openness and financial development literature, we believe that its inclusion can alter our findings. If wide educational gaps which are observed between advanced and developing countries affect the link between financial openness and financial development, then the exclusion of such a variable would certainly introduce a measurement bias. Following the examples of the educational attainment indicators from the economic growth literature we use secondary school gross enrollment rate as a control for educational attainment.

Lastly we employ four *legal and institutional measures* to control for institutional, legal, political and economic factors that may affect the overall level of financial development. These indicators are constructed using subjective, perceptions-based data that reflect views of respondents, agencies and organizations. We use *government effectiveness, regulatory quality, rule of law,* and *control corruption* as measures of legal and institutional quality. These variables do not fluctuate widely across time and are measured in a range from -2.5 to 2.5 where higher values correspond to better governance outcomes.⁵ Given that our analysis is based on panel data specifications that vary across time, the use of almost time-invariant control variables may constitute a main drawback. However, as Chinn and Ito (2006) explain, the inclusion of these time-invariant factors do not pose a substantial problem for our analysis since the characteristics of the institutional quality variables are likely to change very slowly (Chinn and Ito, 2006). On this note, due to the relatively small fluctuation structure of these indicators, we take averages of two consecutive years to replace the missing years' data in the *Worldwide Governance Indicators* database for four legal/institutional quality variables.

3. Aggregate Indices

By combining different measures of financial openness and financial development into a single index we summarize the comprehensive nature of the financial sector. We describe four different indexing techniques below.

3.1 Equally weighted indices

We construct equally weighted indices for financial openness, banking sector, bond and stock market, and overall financial development as well as for institutional quality. The biggest concern with equally weighted indicators is that the weight structure may over-or-underestimate the importance of such measures which could potentially bias our results. In order to avoid this possibility we construct indices using other approaches.

3.2 Coefficient of variation type indices

The second methodology uses the coefficient of variation approach. The weights for financial openness, financial development, and institutional quality are calculated using the coefficient of variation for each variable and the sum of all coefficients of variation for all variables to be used in the index. Weights following this method are constructed as follows:

⁴ Chinn and Ito (2006) argue that logarithm of GDP per capita is important in accentuating the link between financial deepening and rising income levels.

⁵ Please refer to the Worldwide Governance Indicators (WGI) dataset. 1996 – 2009. World Bank. <u>http://info.worldbank.org/governance/wgi/index.aspx#home</u>

$$w_i = \frac{cv_i}{scv}$$

where scv is the sum of all coefficients of variations for variables, and cv_i is the coefficient of variation for each variable *i*, and is calculated as:

$$cv_i = \frac{\sigma_i}{\mu_i}$$

 μ_i and σ_i in the above equation represent the mean and standard deviations of each variable *i*. This procedure allows for weighing each variable differently in the financial openness, financial development and institutional quality indices. It thereby helps avoid the potential bias that may occur when using equal weights.

We can construct an index for financial openness as:

Financial openness_{i,t} =
$$\sum_{i=1}^{3} w_i (meas)_{i,t}$$

where w_i is the relative weight of each variable in the financial openness index and $meas_i$ denotes each of the measures used in constructing the financial openness index. $meas_i$ represents foreign direct investment (% of GDP), international debt issues (% of GDP) and portfolio investment flows (% of GDP). Similarly, banking sector, bond and stock market and financial development, and institutional quality indices are constructed as weighted averages of the corresponding variables following this methodology.

3.3 Principal component analysis for creating indices

Principal components analysis in its simplest form involves a mathematical procedure that helps transform a number of possibly correlated variables into a smaller number of uncorrelated ones.⁶ This method has been shown to be more efficient in establishing optimal weights of variables in comparison to other methods where variables are given equal or subjective weights. One major problem in using principal component analysis is to decide how many components to retain. Four different criterions are suggested in the literature; eigenvalue-one criterion, the scree test, proportion of the variance accounted for by each component, and the interpretability criteria. In this analysis we use the first component which accounts for the maximal amount of variance among the observed variables (Hatcher, 1994). However, in order to avoid particular bias, we suggest a principal component method which relies on information from all components.

We construct principal components indices for financial openness, financial development and institutional quality. For financial development we construct three sub-indices; the banking sector development index, the stock market development index and the bond market development index.

(a) First score principal component índices

To construct our principal component type indices using the first approach we score the first principal component of three individual measures of financial openness. Similarly the index of financial development will be determined by the first principal component of a total of ten variables. We construct first principal component indices for banking sector, stock and bond market development. Institutional quality index is constructed using the first score of the four variables; government effectiveness, regulatory quality, rule of law and control of corruption.

(b) Principal components indices that account for information from all components

We use an additional principal component index that takes into account all possible components. By doing so, we do not discard any information that could potentially affect our estimations. This principal component index proposed by

⁶ For a more in depth discussion of the principal component analysis please refer to Jackson (1991), Dunteman (1989) and Jolliffe (2002).

Bo and Woo (2008) offers a method for calculating weights for individual measures.⁷ According to this methodology the weights for each measure of the index are constructed as follows:

$$w_j = \frac{\sum_{i=1}^{i=p} \lambda_i \alpha_j^i}{\sum_{i=1}^{i=p} \lambda_i}$$

where λ_i (*i* =1,...,*p*) is the *i*th eigenvalue and α_{px1}^i (*i* = 1, ..., *p*) is the *i*th eigenvector of the correlation matrix R_{pxp} respectively (Bo and Woo, 2008). Supposing that $\lambda_1 > \lambda_2 > \lambda_3 > ... > \lambda_p$ and denoting the *i*th principal component as *PC_i* we obtain:

$$PC_i = X\alpha^i$$

where X represents a multi-dimensional matrix that is compromised of normalized transformations of the variables it includes and:

$$\lambda_i = var(PC_i)$$

This implies that the first principal component is the linear combination of the initial indicators, and has the largest variance (Bo and Woo, 2008). The second principal component has the second largest variance and is a linear combination of the indicators which is orthogonal to the first principal component, and the *p*th principal component is a linear combination of the indicators and has the smallest variance.

The index is constructed taking into account the relative importance of all indicators:

$$Index = \frac{\sum_{i=1}^{i=p} \lambda_i PC_i}{\sum_{i=1}^{i=p} \lambda_i} = \frac{\sum_{i=1}^{i=p} \sum_{j=1}^{i=p} \lambda_i \alpha_j^i x_j}{\sum_{i=1}^{i=p} \lambda_i} = \sum_{j=1}^{j=p} w_j x_j$$

where x_j (j = 1, ..., p) is the *j*th column of the matrix X and w_j is the final weight of the indicator *j*. All variables that constitute the *j*th column of the matrix X, x_j , are standardized. The sum of the weights expressed by the formula above does not necessarily have to equal unity. This is due to the fact that the principal component analysis in its underlining structure normalizes the mode of each eigenvector to unity. The weights therefore could be very close to but not always equal to 1 (Bo and Woo, 2008).

Following this methodology and taking into account information from all components, we construct indices with standardized individual measures for financial openness, banking sector, stock and bond market development, financial development and institutional quality.⁸ This avoids any potential problem that could arise as a result of different scales or units of measurement.

4. Empirical Model

Various estimation techniques have been employed in literature for the examination of the link between financial openness and financial development. While Chinn and Ito (2006) utilize simple, point estimate OLS, Huang (2006) argues for fixed effects estimation with levels, OLS estimation with first differences, and Arellano-Bond dynamic panel data model with GMM estimators. Baltagi, Demetriades, and Law (2007) and Demetriades and Law (2006) find dynamic panel data models more appropriate. Dealing with unobserved heterogeneity one can refer to the within-group fixed effects and first differences estimators. Unfortunately neither estimator captures the partial adjustment property that accounts for the new information that explanatory variables that carry out (Huang, 2006). To account for this property and obtain consistency in estimations we employ the dynamic Arellano-Bond GMM panel data procedure to overcome

⁷ The principal components methodology of Bo and Woo (2008) is similar to that proposed by the United Nations Conference on Trade and Development (UNCTAD) for constructing the Trade and Development Index. Bo and Woo (2008/2010), Nagar and Basu (2002) and Klein and Ozmucur (2002/2003) provide different approaches in constructing indices analogous to the Trade and Development Index (TDI) with minor alterations.

⁸ This method makes use of all eigenvectors and proposes to use weights depending on the eigenvectors and eigenvalues.

the Nickell bias that occurs when the lagged dependent variable is correlated with the disturbance. Ultimately our Arellano-Bond dynamic panel data model of 61 countries and twelve years can be represented as:

$$\Delta FD_{it} = \sum_{k=1}^{n} \gamma_k \, \Delta FD_{it-k} + \beta \Delta FO_{it} + \sum_{j=1}^{s} \lambda_j \, \Delta Control_{jit} + \Delta \varepsilon_{it}$$

where γ_k measures the speed of adjustment, β denotes the short-run effect of financial openness on financial development which we expect to be positive, and $Control_{jit}$ is the group of control variables of trade openness, logarithm of GDP per capita, secondary school enrollment rate, and institutional quality.⁹

The above model no longer has a country specific effect. The Arellano-Bond dynamic panel data model accounts for the individual effects. All terms in the above model have been differenced in order to elude the Nickell bias. Moment conditions require that:

$$E[FD_{it-k}\Delta\varepsilon_{it}] = 0$$
 for $\forall k \ge n+1$

This condition guarantees the lagged dependent variable to be uncorrelated with the first difference of the error terms although the first difference of the lagged dependent variable could easily be correlated with the first difference of the error terms.

Two diagnostic tests for serial correlation are derived by the model. We expect to find first order serial correlation in first differenced residuals because $\Delta \varepsilon_{it}$ and $\Delta \varepsilon_{it-1}$ contain the same term, ε_{it-1} . Second order and higher serial correlations could create further problems due to the difficulty in verifying the validity of the moment conditions.¹⁰

Given our model background, following Baltagi, Demetriades, and Law (2007), we test for the following hypotheses:

- (a) Do trade and financial openness influence financial development? What are the effects of economic and legal institutions on financial development over and above the effects of openness? Do educational indicators affect financial development?
- (b) Does simultaneous opening of both financial and trade accounts bring additional benefits to financial development?

Following these two hypotheses, we specify the following dynamic equations for financial development:

Model (a): (Without an interaction term)

$$\Delta FD_{it} = \gamma \Delta FD_{it-1} + \beta \Delta FO_{it} + \lambda_1 \Delta TO_{it} + \lambda_2 \Delta log GDPpc_{it} + \lambda_3 \Delta INSQUA_{it} + \lambda_4 \Delta SECSCHOOL_{it} + \Delta \varepsilon_{it}$$
(1.1)

Model (b): (With an interaction term)

$$\Delta FD_{it} = \gamma \Delta FD_{it-1} + \beta \Delta FO_{it} + \lambda_1 \Delta TO_{it} + \lambda_2 \Delta log GDPpc_{it} + \lambda_3 \Delta INSQUA_{it} + \lambda_4 \Delta SECSCHOOL_{it} + \lambda_5 \{\Delta FO_{it} \times \Delta TO_{it}\} + \Delta \varepsilon_{it}$$

$$(1.2)$$

⁹ Any influence of financial openness is now conditioned on the history controlled by the first differenced lagged dependent variable (Huang, 2006). This model is a restricted version of the static fixed effects specification and it includes dynamic effects through the lagged dependent variable that is included as a right hand side regressor.

¹⁰ To test for overidentifying restrictions we perform Sargan tests. We do not include the lag of the dependent variable as an instrument. We utilize the Arellano-Bond two-step estimator to avoid any panel-specific autocorrelation and to obtain better diagnostics results.

where *FD*_{*it*} is the financial development index, *FO*_{*it*} is the financial openness index, *TO*_{*it*} is trade openness, *logGDPpc*_{*it*} is the natural logarithm of GDP per capita, *INSQUA*_{*it*} is the institutional quality index and *secschool*_{*it*} is the secondary school enrollment rate.

The two models express the link between financial openness and financial development and examine whether a simultaneous effect of financial and trade openness leads to additional benefits in terms of financial development. In the first model, Model (a), we would expect to have positive and significant coefficients for both β and λ_1 . Similarly if λ_3 and λ_4 are found to be positive then, improvements in institutions and the quality of legal systems and education will enhance financial development. To test for the second model, Model (b), we introduce an interaction term which represents the significance of opening both markets at the same time. If λ_5 is found to be positive, we can state that the simultaneity hypothesis between financial and trade openness has additional benefits for financial development.

Both models serve to advance the literature in terms of the methods used in the estimation process. Given our objective of establishing the importance of financial development as an element influencing economic growth and welfare, and given that the literature has yet to find an answer to why we observe differences among countries in extracting the benefits of financial liberalization, our analysis stands as a starting point for determining the transition mechanism between financial openness and financial development.¹¹

There remain to be a few drawbacks to our estimations. The literature shows that in many cross-sectional studies both developing and advanced countries, are lumped together into the same sample. As Henry (2006) explains, including both sets of countries increases the sample size and could lead to more efficient results in estimation. However, doing so without employing an empirical methodology, which particularly recognizes the fundamental theoretical difference between advanced and developing countries, would undermine the study's ability to interpret the data. In order to correct for the problem, we first estimate our model with a full sample and then subgroup the data into developing and advanced countries so as to compare the results obtained in both estimation procedures.

5. Empirical Results

We discuss the results of the dynamic panel data model.¹² Overall the results demonstrate that all indices have high correlations with each other while the banking sector development index has higher correlations with the financial development index.

5.1 Results using equally weighted indices

Our empirical estimations for equally weighted indices are presented in Table 1 (a).¹³ In the benchmark dynamic GMM estimations, all variables other than the lags of the dependent variable are treated as exogenous. The four dependent variables used in our regressions are banking sector, bond and stock market, and financial development indices. Table 1 (a) shows that the equally weighted financial openness index is statistically significant for three columns. Financial openness has a positive coefficient for all dependent variables with the exception of the equally weighted bond market index.

The rest of the results from Table 1 (a) indicate that trade openness is positive and significant for all development indices, whereas logarithm of GDP per capita is significant and positive for all dependent variables with the exception of the equally weighted bond market index. The equally weighted institutional quality index is negatively significant for three dependent variables. This finding contravenes the one of the literature which states that the effects of higher

 ¹¹ Our indexing strategies allow the data to tell us how to measure financial openness and financial development. Our index measures help overcome the problem of the particular use/choice of individual measures that exists in the literature.
 ¹² Table 1 in the Supplementary Appendix gives a brief summary of the variables used in our estimation procedure. Tables 2 (a) to (d) depict the relations of the indices and the control variables. The Supplementary Appendix can be found at: https://docs.google.com/viewer?a=v&pid=sites&srcid=ZGVmYXVsdGRvbWFpbnx6ZXluZXBvemtva3xneDo2ZTA3YmI0NDU4YmE5ZDV

¹³ In the benchmark dynamic GMM estimations, all variables other than the lags of the dependent variable are treated as exogenous.

development in terms of institutions should be carried out to all sources of financial development. Baltagi, Demetriades and Law (2007) using individual dependent variables find institutional quality to be positive whenever significant but negative when the stock market capitalization is used as the dependent variable. Similarly in Chinn and Ito (2006) institutional quality has altering signs depending on the measures chosen and the dependent variables to be used. Our finding could therefore reinforce the argument that a threshold level of institutional and legal development is needed in order for countries to benefit from financial openness. One possible explanation is that due to the dynamic structure of our model, our institutional quality indices depict changes of institutional differences. This then implies that if countries have passed a certain level of institutional development any further deviations may have negative impacts on overall level of banking sector, bond market, stock market and financial development.

The results of the diagnostic tests show that in all cases the first order serial autocorrelation is rejected and the second order is accepted. The Sargan test, on the other hand, cannot reject the null hypothesis in all cases. Our results are in line with those of the literature. Our initial findings highlight the link between financial openness and bond market development.

5.2 Results using coefficient of variation type indices

Table 1 (b) finds financial openness index to be significant for all dependent variables however the coefficients are positive only when the stock market development and financial development indices are used as dependent variables. Trade openness is shown to be positively significant for all indices. Logarithm of GDP per capita is positive and significant for all cases with the exception of the bond market development index. Secondary school enrollment rate enters with a negatively significant coefficient for all cases with the exception of the bond market development index, whereas the institutional quality index is negative and significant for all cases. Similarly the first order serial correlation is rejected in all cases and the Sargan test cannot reject the null hypothesis of over-identification for all cases.

5.3 Results using principal component analysis type indices

Results using first score principal components¹⁴

The results of the principal component analysis of financial openness, banking sector development, bond market development, stock market development, overall financial development and institutional quality depict that the first component captures a large percentage of the total variation of individual measures represented by the eigenvectors.¹⁵ The analysis demonstrates similar weights for individual variables of financial openness and financial development. Using any single measure to study the impact of financial openness on financial development would therefore bias our results.¹⁶ Nevertheless we use the first principal component to score a proper and an efficient indices for financial openness, financial development and institutional quality.

Table 2 (a) depicts our findings for first score principal component indices. The results show that the financial openness index is significant in all cases with the exception of the bond market development index. Likewise trade openness is positive and significant for all dependent variables. This could mean that trade openness affects the development of bond markets whereas financial openness does not have a significant effect on the overall enhancement of bond markets.

Control variables of institutional quality and secondary school enrollment rate are mostly significant however with differing magnitudes. Secondary school enrollment rate enters with a positive coefficient in two out of four cases, whereas institutional quality is found to be negatively significant for all cases. The diagnostics are satisfactory for all

¹⁴ We verify that the individual variables used in our indices are correlated. Our results show that we have positive correlations among individual variables.

¹⁵ The results can be found in the Supplementary Appendix.

¹⁶ The examination of the eigenvectors portrays that the use of individual variables to examine the link between financial openness and financial development would potentially lead to selection bias.

dependent variables used. The absence of first order serial correlation is rejected and the Sargan test results cannot reject the over-identification restrictions.¹⁷

Overall, our findings show that there is a significant link between financial openness and financial development when both measures are constructed using first score of principal components. The results also demonstrate that the link between financial openness and financial development becomes more comprehensible with first score indices in comparison to equally weighted and coefficient of variation type indices.¹⁸

Results using principal components that take into account information from all components

The results in Table 2 (b) indicate that the financial openness index is significant for all dependent variables. Trade openness is also positively significant for all cases. Logarithm of GDP per capita is significant for all cases with the exception of the bond market development index. First order serial autocorrelation tests are rejected, while the Sargan test of over-identification cannot reject the null hypothesis in all cases.

Our results convey that using further components in our indices depicts similar findings for the link between financial openness and financial development. Due to the similarity of the results with both types of principal component indices, it is difficult to identify one methodology as the ideal procedure in constructing indices. In this context we choose the index measure that maximizes the goodness of fit of the linear model between financial openness and financial development. Further examination of Tables 2 (a) and (b) conveys that while first score indices depict higher significances in terms of t-statistics, this methodology of principal components provides better results in terms of magnitudes and interpretations. We thereby use this methodology in reporting the rest of our results.

5.4 Adding the interaction term

Findings from Table 3 which include the interaction term show that the effects of financial and trade openness indices are positive and significant for all dependent variables. Logarithm of GDP per capita is positive and significant for all dependent variables with the exception of the bond market development index. The institutional quality index is negatively significant for banking sector, bond and stock market development and positively significant for financial development whereas secondary school enrollment rate is positively significant for three dependent variables. The interaction term between financial and trade openness is significant in three out of four cases with changing magnitudes. It is positive and significant for the banking sector development index, whereas it is negative and significant for the banking sector development index, whereas it is negative and significant for bond and stock market development indices. This finding implies that a simultaneous opening of financial and goods markets leads to a decline in bond and stock market development rates. This could potentially be the result of the effects of financial openness and trade openness being picked up separately by each variable and not by the interaction term. Our results however agree with those of the literature. Baltagi, Demetriades and Law (2007) find the interaction term to be statistically significant together with financial and trade openness.¹⁹ Law and Demetriades (2006) find the interaction term, trade openness and capital account openness to be positively significant for stock market development.

6. Robustness checks and further issues

We carry out a large number of robustness checks in order to examine the sensitivity of our results to alternative specifications and methods. Here, we only report some robustness checks. Below we discuss the results when heterogeneity is taken under examination.

6.1 Robustness checks

¹⁷ Huang (2006) reports similar results to our initial estimations. He finds the financial openness index constructed using the first principal components to be positive and significant for stock market, banking sector and financial development.

¹⁸ One pitfall is that the link between financial openness and bond market development disappears as in the case with equally weighted indices.

¹⁹ Using banking sector development measures as dependent variables, the authors find the interaction term to be significant and negative for private credit, liquid liabilities and domestic credit.

The first set of robustness examines whether the exclusion of control variables of secondary school enrollment rate and institutional quality index affect our proposed model. The exclusion of institutional quality index from our regressions, not reported here²⁰, does not affect our initial findings. Similarly exclusion of the secondary school enrollment rate causes trade openness to lose significance for some dependent variables. These results show that even though the exclusion of institutional quality does not remarkably change our results, the exclusion of secondary school enrollment rate does. We thereby, stress the importance of including both of these control variables in our analysis to examine the link between financial openness and financial development.

Further tests regarding the structure of the variables have been considered. In order to account for the fact causality we run robustness checks to examine the simultaneity and the endogeneity issues. Treating all right hand side variables as predetermined does not alter our main findings. With predetermined variables we find that trade openness enters with a negative coefficient when banking sector and bond market development indices are used as dependent variables. Similarly treating financial and trade openness as predetermined variables does not change our findings. Treating openness variables as endogenous we find similar results to those reported in Table 2 (b). This thereby confirms that the assumption of treating right hand side regressors as exogenous variables is not a restrictive one for this analysis.

Another important issue to be analyzed is the time series versus cross sectional effects. Due to the twelve year period selected in our model we believe that there may be time series effects that are not picked up by our one lag structure. In order to account for these effects we estimate a higher lag model to observe the sensitivity of our findings to different lag structures. Table 4 depicts the results when three lags of the dependent variable are used as right hand side regressors and three lags of variables explanatory variables are used as instruments. The results show that adding further lags does not change our main findings. With further lags we find trade openness to be insignificant for banking sector development index and negatively significant for the bond market development index.²¹ Our analysis with higher lags demonstrates that the qualitative nature of our results is robust to alternative specifications and estimation methods.

6.2 Further issues

Heterogeneity problems

The use of developing and advanced countries together can create a drawback to our estimated model. Even though having a large number of observations in a joint sample leads to more efficient results we may not be able to fully differentiate whether the effects of financial openness on financial development follow due to the influence of advanced economies. In order to avoid complications and to clarify issues on heterogeneity we split our data set into developing and advanced countries and examine the results when the developing country sample is used in estimations.²²

The developing country sample includes 31 countries over the twelve year period. The findings from Table 5 show that financial openness is positive and significant for banking sector, stock and bond market and financial development indices. The results are similar to those reported in Table 2 (b) in terms of sign and significance; however, trade openness, institutional quality and secondary school enrollment depict different magnitudes of coefficients. The results, overall, confirm that we do not have a heterogeneity problem that is not explicitly recognized by our estimation methods. The link between financial openness and financial development follows even when a subsample of developing countries is used in our analysis. This strengthens the argument regarding the importance of opening up financial markets so as to enhance banking sector, bond and stock markets in developing countries.

Additional results

To compare our results with earlier studies we run a series of regressions. To provide a better understanding of the effectiveness of our de facto type principal component indices we first present our results with de jure type capital

²⁰ Please refer to the Supplementary Appendix.

²¹ Other lag specifications do not change our results.

²² We rely on the World Bank's income group definition when splitting our data set into developing and advanced countries.

account openness index of Chinn and Ito. Chinn and Ito's capital account openness index is constructed using four binary dummy variables which represent restrictions on financial transactions across countries as reported in the IMF's Annual Report on Exchange Arrangements and Exchange Restrictions (Baltagi, Demetriades and Law, 2009). The four binary variables (existence of multiple exchange rates, restrictions on current account, capital account transactions, and a variable for export proceeds) are summarized in an index using the first score of principal components (Chinn and Ito, 2006). Table 6 depicts results with our principal component type development indices and the Chinn and Ito index of capital account openness. Capital account openness index is found to be positively significant for banking sector and stock market development indices and negatively significant for bond market and financial development indices. These findings, which are different from our benchmark model of Table 2 (b), imply that the removal of restrictions from capital accounts and the allowance of capital movements across countries have a negative impact on bond market and overall financial development. It is interesting to note that the diagnostics fail to compete with the ones reported in Table 2 (b). In three out of four cases first order serial correlation is rejected and the Sargan test cannot reject the null hypothesis of over-identification although the Sargan test values remain to be less powerful in comparison to those of the benchmark case.

In Tables 7 and 8 we compare our previous results to cases where a de jure type financial openness measure (the Chinn and Ito capital account openness index) and our financial openness index is regressed against most frequently used individual financial development measures. Once again the findings demonstrate that our principal component index of financial openness provides more significant and economically profound results in comparison to the Chinn and Ito index. Yet, in both cases the findings for the first order serial correlation test and the Sargan test are weaker in contrast to those from Table 2 (b). Although the use of de jure type measures may be limiting in terms of interpretations, they help in resolving issues related to the simultaneity hypothesis. The results using the discrete de jure type financial openness index shows that the link goes in the direction of financial openness to financial development. Given that our index measures are stronger in terms of robustness, statistical significances and interpretations in comparison to the de facto measures which help resolve causality issues, we believe our indices also carry this property. Our previous results using endogenous financial openness indices depict similar findings to our benchmark regressions providing evidence that a causality issue is not undermining our analysis.

Additionally we estimate our model with individual financial openness and financial development measures. Our findings suggest mixed results.²³ Depending on the choice of financial development measures, the three indicators of financial openness; foreign direct investment (% of GDP), international debt issues (% of GDP) and portfolio investment (% of GDP) alter signs.

Lastly we analyze the results when individual financial openness measures of foreign direct investment, international debt issues and portfolio investment are regressed against principal component type indices for financial development. Our findings support our earlier argument on the choice of financial openness measures affecting the results. Using individual measures as dependent variables for financial development, or using individual variables as measures of financial openness (both de facto and de jure type individual measures) we cannot find the clear link that we observe between financial openness and financial development whilst using indices. The results clearly demonstrate that depending on the variables chosen, the relationship between financial openness and financial development alters. Nevertheless using principal components indices we can unmistakably find a positive link between financial openness and financial openness and financial development set.

Partial effects and possible policy implications

To calculate the total effect of opening up goods and financial markets we use partial derivatives of banking sector, bond and stock market and overall financial development with respect to trade and financial openness. Recalling Model (b) in equation (2) we evaluate the partial derivates of financial development with respect to trade and financial openness as:

²³ The results are reported in the Supplementary Appendix. Comparisons to the literature should be taken with caution due to a variety of reasons. Both models used in estimations, the choice of years and countries widely differ across studies.

$$\frac{\partial FD_{it}}{\partial TO_{it}} = \lambda_1 + \lambda_5 FO_{it}$$
$$\frac{\partial FD_{it}}{\partial FO_{it}} = \lambda_1 + \lambda_5 TO_{it}$$

Examining the results in Table 3 the partial derivative of the financial development index with respect to trade openness at the mean level of financial openness index is 0.006.²⁴ Alternatively at the mean level of trade openness, the partial derivative of financial development index with respect to financial openness index is 0.482.

Our findings show that both partial derivatives are positive even for the most closed economies implying that opening both trade and financial markets may have a large impact on overall financial development. However bond and stock market development indices have negative coefficients for the interaction term of financial and trade openness. Examining the partial derivatives for both bond and stock market development indices we find positive values.²⁵ Our results thereby suggest that a simultaneous opening could be beneficial in terms of banking sector, bond and stock markets and overall financial development. Nevertheless, possible implications of these results should be taken with great precaution.

We have shown that our principal component financial openness index influences banking sector, bond and stock market and overall financial development. We have demonstrated a clear positive link between financial openness and financial development using different lag structures, time dummies, models, and specifications. However, the results for the coefficients reported so far all represent short-run effects due to the presence of the lagged dependent variable in the models. In order to obtain long-run effects particularly for the financial openness index in Table 2 (b) we divide financial openness coefficients by $1 - \gamma$, where γ is the coefficient of the lagged dependent variable from our regressions. The findings show that the link between financial openness and development indices is positive even when examining the long-run coefficients.²⁶ These results suggest that financial openness has a clear impact on long-run financial development.

7. Concluding remarks

The literature has broadly focused on the link between financial liberalization and economic growth nevertheless the relationship between financial liberalization and financial development lacked a thorough discussion. Our goal has been to construct comprehensive indicators for both financial openness and financial development and analyze this link.

Using a panel data set of developing and advanced countries our results with aggregate indices of equal weights, coefficient of variation and two different principal component type methodologies show that the relationship between financial openness and financial development exists and that it is more significant than what the literature has previously acclaimed. Further examination of our benchmark results depict that the principal component indices with information from all components provide higher significances and more meaningful interpretations. In all estimations, independent of the development variables chosen, the principal component index of financial openness is found to be positive and significant. The inclusion of an interaction term, to analyze whether the simultaneous opening of financial and goods markets have additional effects on financial development, brings intriguing results. Our findings show that the interaction term is positively significant for the banking sector development index and negatively significant for bond and stock market development indices. By splitting the sample we confirm that the link between financial openness and financial development exists even for developing countries.

²⁴ The same partial derivative evaluated at the minimum level of financial openness index is 0.005, and evaluated at the maximum value is 0.006. Alternatively at the minimum level of trade openness, the partial derivative of financial development index with respect to financial openness index is 0.48 and it is 0.489 at the maximum level of trade openness.

²⁵ Calculations are available upon request.

²⁶ The long-run coefficient for financial openness using the banking sector development index is 0.643. The coefficients for financial openness using the bond and stock market development indices are 0.505 and 0.107 whereas for the overall financial development index, it is 0.715.

Comparisons with the literature stress the strength and effectiveness of our indices. Principal component type indices provide more robust and stronger statistical conclusions on the link between financial openness and financial development. The partial derivatives and the long-run coefficients convey the positive link between financial openness and financial development validating our benchmark results.

On a last note, we have shown that using indices the link between financial openness and financial development is unambiguous. To this end, future work would call for a model that can provide theoretical underpinnings for the effects of financial openness on financial development.

Acknowledgements

This paper was previously circulated under the name "Breaking the misconceptions: An analysis of the effects of financial openness on financial development using index measures". I am grateful to Klaus Desmet for his invaluable help. I further thank Juan José Dolado, Matthias Kredler, Ricardo Mora, Carlos Velasco and Ulrich Wagner for useful comments and suggestions.

References

- Alfaro, L., S. Kalemli-Ozcan, and V. Volosovych. 2005. "Capital Flows in a Globalized World: The Role of Policies and Institutions." NBER Working Paper 11696
- Arestis, P., and P. O. Demetriades. 1997. "Financial Development and Economic Growth: Assessing the Evidence." Economic Journal, 107 (442): 783 – 799
- Aziakpono, J. M. 2007. "Effects of Financial Integration on Financial Development and Economic Performance of the SACU Countries." *Paper presented at the ECA/ADB African Economic Conference*, Addis Ababa, Ethiopia
- Baltagi, B., P. Demetriades, and S. H. Law. 2007. "Financial Development, Openness, and Institutions: Evidence from Panel Data." *Paper presented at the IMF Conference on New Perspectives on Financial Globalization*, Washington, DC
- Baltagi, B., P. Demetriades, and S. H. Law. 2009 "Financial Development, Openness, and Institutions: Evidence from Panel Data." *Journal of Development Economics*, 89 (2): 285 296
- Beck, T., A. Demirgüç-Kunt, and R. Levine. 2000. "A New Database on Financial Development and Structure." World Bank Economic Review 14, 597 – 605
 <u>http://econ.worldbank.org/WBSITE/EXTERNAL/EXTDEC/EXTRESEARCH/0,,contentMDK:20696167~pagePK:642</u> <u>14825~piPK:64214943~theSitePK:469382,00.html</u> (Accessed: March 7, 2014)
- Beck, T., A. Demirguc-Kunt, and R. Levine. 2009. "Financial Institutions and Markets across Countries and Over Time Data and Analysis." World Bank Policy Research Working Paper WPS 4943
- Bo, C., and Y. P. Woo. 2008. "A Composite Index of Economic Integration in the Asia-Pacific Region." Asia Pacific Foundation of Canada Working Paper
- Bo, C., and Y. P. Woo. 2010. "Measuring Economic Integration in the Asia-Pacific Region: A Principal Components Approach." Asian Economic Papers, 9 (2): 121 – 143
- Bond, S. R., A. Hoeffler, and J. Temple. 2001. "GMM Estimation of Empirical Growth Models." University of Oxford Economics Papers 2001-W21
- Bussiere, M., and M. Fratzscher. 2004. "Financial Openness and Growth: Short-run Gain, Long-run Pain?" European Central Bank Working Paper Series 348
- Chinn, M., and H. Ito. 2002. "Capital Account Liberalization, Institutions and Financial Development: Cross Country Evidence." NBER Working Paper 8967
- Chinn, M., and H. Ito. 2006. "What Matters for Financial Development? Capital Controls, Institutions and Interactions." Journal of Development Economics, 81 (1): 163 – 192
- Chinn, M., and H. Ito. 2008. "A New Measure of Financial Openness." *Journal of Comparative Policy Analysis*, 10 (3): 309 322. <u>http://web.pdx.edu/~ito/Chinn-Ito_website.htm</u> (Accessed: March 7, 2014)
- De Gregorio, J. 1999. "Financial Integration, Financial Development, and Economic Growth." <u>http://www.econ.uchile.cl/uploads/publicacion/c6b5fc4b-498f-462e-9453-2ae0325e0b2f.pdf</u> (Accessed: March 7, 2014)
- Demetriades, P. O., and S. Andrianova. 2005. "Sources and Effectiveness of Financial Development: What We Know and What We Need to Know." UNU – WIDER World Institute for Development Economic Research Working Papers RP2005/76

- Demetriades, P. O., and S. H. Law. 2004 "Capital inflows, trade openness and financial development in developing countries." *Money Macro and Finance (MMF) Research Group Conference 38*
- Demetriades, P. O., and S. H. Law. 2005. "Finance, Institutions, and Economic Growth." University of Leicester Department of Economics Discussion Paper 04/5
- Demetriades, P. O., and S. H. Law. 2006. "Openness, Institutions and Financial Development." WEF Working Papers 0012
- Demirguc-Kunt, A., and R. Levine. 2001. *Financial Structure and Economic Growth: A Cross-Country Comparison of Banks, Markets, and Development*. Cambridge, Massachusetts: MIT Press
- Dunteman, G. H. 1989. Principal Components Analysis. Newbury Park, California: Sage Publications
- Edison, H., R. Levine, L. Ricci, and T. Slok. 2002. "International Financial Integration and Economic Growth." *Journal of International Monetary and Finance*, 21 (6): 749 – 776
- Edwards, S. 2004. "Financial Openness, Sudden Stops, and Current Account Reversals." *American Economic Review*, 94 (2): 59 64
- Hatcher, L. 1994. A Step-by-Step Approach to Using SAS for Factor Analysis and Structural Equation Modeling. Cary, North Carolina: SAS Institute Inc.
- Henry, P. 2003. "Capital Account Liberalization, the Cost of Capital, and Economic Growth." *American Economic Review*, 93 (2): 91 96
- Henry, P. 2006. "Capital Account Liberalization: Theory, Evidence, and Speculation." NBER Working Paper 12698
- Huang, W. 2006. "Emerging Markets, Financial Openness, and Financial Development." University of Bristol Economics Discussion Papers 06/588
- Ito, H. 2006. "Financial development and financial liberalization in Asia: Thresholds, institutions and the sequence of liberalization." *The North American Journal of Economics and Finance*, 17 (3): 303 327
- Jackson, E. 1991. A User's Guide to Principal Components, New Jersey: John Wiley & Sons, Inc.
- Jolliffe, I.T. 2002. Principal Component Analysis, New York: Springer-Verlag, Inc.
- Judson, R. A., and A. L. Owen. 1996. "Estimating Dynamic Panel Data Models: A Practical Guide for Macroeconomists." Federal Reserve Board of Governors Finance and Economics Discussion Series 1997-3
- Kaminsky, G. and S. Schmukler. 2003. "Short-Run Pain, Long-Run Gain: The Effects of Financial Liberalization." NBER Working Paper 9787
- Kaufmann, D., A. Kraay and M. Mastruzzi. 2009. "Governance Matters VIII: Governance Indicators for 1996-2008." World Bank Policy Research Working Paper Series 4978
- Kaufmann, D., A. Kraay and M. Mastruzzi. 2010. "The Worldwide Governance Indicators: Methodology and Analytical Issues." World Bank Policy Research 5430
- Klein, M., and G. Olivei. 1999. "Capital Account Liberalization, Financial Depth, and Economic Growth." *Federal Reserve* Bank of Boston Working Papers 99-6
- Klein, L. R., and S. Ozmucur. 2002/2003. "The Estimation of China's Economic Growth." Journal of Economic and Social Measurement, 67 (8): 187 – 202

- Kose, M. Ayhan, E. S. Prasad, K. S. Rogoff, and S.-J. Wei. 2003. "The Effects of Financial Globalization on Developing Countries: Some Empirical Evidence." *IMF Occasional Paper 220*
- Kose, M. A., E. S. Prasad, K. S. Rogoff, and S.-J. Wei. 2006. "Financial Globalization: A Reappraisal." *IMF Staff Papers* Palgrave Macmillan Journals 56 (1)
- Lane, P. R., and G. M. Milesi-Ferretti. 2003. "International Financial Integration." IMF Working Paper WP 03/86
- Law, S. H., and W.N.W Azman Saini. 2008. "The Quality of Institutions and Financial Development." MPRA Paper 12107
- Levine, R. 2001. "International Financial Liberalization and Economic Growth." *Review of International Economics*, 9 (4): 684 698

Mody, A., and A. P. Murshid. 2005. "Growing Up with Capital Flows." *Journal of International Economics*, 65 (1): 249 – 266

- Nagar, A. L., and S. R. Basu. 2002. "Weighting Socioeconomic Indicators of Human Development: A Latent Variable Approach." In Ullah, A. et al. (eds) Handbook of Applied Econometrics and Statistical Inference. Marcel Dekker, New York
- Nissanke, M., and H. Stein. 2003. "Financial Globalization and Economic Development: Toward an Institutional Foundation." *Eastern Economic Journal*, 29 (2): 287 – 308
- Obstfeld, M. 2008. "International Finance and Growth in Developing Countries: What Have We Learned." Commission on Growth and Development Working Paper 34
- Quinn, D. 1997. "The Correlates of Change in International Financial Regulation." American Political Science Review, 91 (September): 531 551
- Quinn, D., and C. Inclan. 1997. "The Origins of Financial Openness: A Study of Current and Capital Account Liberalization." American Journal of Political Science, 41 (3): 771 – 813
- Quinn, D. 2003. "Capital Account Liberalization and Financial Globalization, 1890 1999: A Synoptic View." International Journal of Finance and Economics, 8 (3): 189 – 204
- Ravallion, M. 2003. "The Debate on Globalization, Poverty and Inequality: Why Measurement Matters." World Bank Policy Research Working Paper Series 3038

Sheret, M. 1984. "The Coefficient of Variation: Weighting Considerations." Social Indicators Research, 15 (3): 289 - 295

- Stulz, R. M. 2006. "Limits of Financial Globalization." The Journal of Finance, 60 (4), 1595 -1638
- Ullah, A., and D. E. A Giles. 1998. Handbook of Applied Economic Statistics, New York: Marcel Dekker, Inc.
- UNCTAD. 2005. "Trade and Development Index." *Developing Countries in International Trade*. <u>http://unctad.org/en/docs/ditctab20051ch1_en.pdf</u> (Accessed: March 7, 2014)
- Williamson, J., and M. Mahar. 1998. "A Survey of Financial Liberalization." Essays in International Finance Princeton University, No. 211
- World Bank. 2007. 2007 World Development Indicators, Washington, D.C.: International Bank for Reconstruction and Development/The World Bank Press
- World Bank and the UNESCO Institute for Statistics. 1970 2009. "EdStats Data Query." The World Bank Group. <u>http://databank.worldbank.org/data/views/variableselection/selectvariables.aspx?source=education-statistics-~-all-indicators</u> (Accessed: January 9, 2014)

World dataBank. 1960 – 2010. "World Development Indicators." The World Bank Group. <u>http://databank.worldbank.org/data/views/variableselection/selectvariables.aspx?source=world-development-indicators</u> (Accessed: January 9, 2014)

Worldwide Governance Indicators. 1996 – 2009. "Aggregate Governance Indicators." The World Bank Group. <u>http://info.worldbank.org/governance/wgi/index.aspx#home</u> (Accessed: January 9, 2014)

Appendix

Table 1 (a): Equally weighted financial openness and financial development indices

		Depende	ent Variables	
	∆Banking sector development	ΔBond market development	ΔStock market development	∆Financial development
∆Dependent variable (Lagged)	0.731***	0.887***	0.376***	0.311***
· · · · · · · · · · · · · · · · · · ·	(269.1)	(76.47)	(451.2)	(42.54)
	[0.00266]	[0.0116]	[0.000834]	[0.00731]
ΔFinancial openness	0.343***	0.005	0.136***	0.801***
·	(31.67)	(0.713)	(21.43)	(21.13)
	[0.0108]	[0.00729]	[0.00635]	[0.0379]
∆Trade Openness	0.024***	0.030***	0.712***	0.186***
	(4.133)	(5.885)	(112.4)	(16.90)
	[0.00575]	[0.00512]	[0.00633]	[0.0110]
ΔLog GDP per capita	3.422***	-3.614***	50.53***	14.24***
	(2.677)	(-3.845)	(36.19)	(17.80)
	[1.278]	[0.940]	[1.396]	[0.800]
ΔInstitutional Quality	-2.040***	-3.841***	-25.76***	0.542
	(-4.783)	(-5.017)	(-54.09)	(0.329)
	[0.427]	[0.766]	[0.476]	[1.650]
ΔSecondary School Enrollment	0.006**	0.008***	-0.239***	-0.041***
	(2.283)	(4.805)	(-16.42)	(-13.24)
	[0.00266]	[0.00171]	[0.0145]	[0.00308]
Number of observations:	333	301	376	262
Number of groups:	50	41	54	38
Arellano – Bond test for	z = -2.07	z = -2.23	z = -1.96	z = -1.82
AR(1) in first differences:	Pr > z = 0.0381	Pr > z = 0.0255	Pr > z = 0.0500	Pr > z = 0.0681
Arellano – Bond test for	z = -1.03	z = -1.48	z = -0.97	z = -1.16
AR(2) in first differences:	Pr > z = 0.3011	Pr > z = 0.1395	Pr > z = 0.3298	Pr > z = 0.2445

Sargan test:	Chi ² (54)	Chi ² (54)	Chi ² (54)	Chi ² (54)
	= 48.43	= 36.50	= 51.02	= 33.02
	Prob > chi ²			
	= 0.6883	= 0.9674	= 0.5901	= 0.9892

Notes: All regressions are estimated using the Arellano – Bond dynamic panel GMM estimation with one lag of the dependent variable to be included in the model, 0 lags of the dependent variable and one lag of other variables to be used as instruments. The results reported here use the twostep estimator. (The estimations use Stata's xtabond command). The financial openness, banking sector development, bond and stock market development, financial development variables and institutional quality measures are constructed using equal weights in indices. Figures in parentheses are t-statistics and the stars represent the significant t statistics for 1%, 5%, and 10% confidence levels respectively. The figures in brackets depict the standard errors for the coefficients.

		Depende	ent Variables	
	ΔBanking sector development	ΔBond market development	ΔStock market development	ΔFinancial development
ΔDependent variable (Lagged)	0.799***	0.883***	0.372***	0.296***
	(99.95) [0.00800]	(62.80) [0.0141]	(538.4) [0.000691]	(62.55) [0.00474]
∆Financial openness	-0.021***	-0.0509***	0.102***	0.749***
	(-3.831)	(-7.189)	(8.597)	(38.82)
	[0.00548]	[0.00708]	[0.0119]	[0.0193]
∆Trade openness	0.104***	0.0415***	0.775***	0.521***
	(17.46)	(5.303)	(115.7)	(31.22)
	[0.00595]	[0.00783]	[0.00670]	[0.0167]
ΔLog GDP per capita	4.875***	-3.427***	53.32***	21.21***
5	(6.258)	(-2.770)	(46.46)	(14.45)
	[0.779]	[1.237]	[1.148]	[1.468]
Δ Institutional Quality	-6.319***	-5.337***	-27.96***	-11.94***
	(-65.43)	(-10.66)	(-49.14)	(-3.424)
	[0.0966]	[0.501]	[0.569]	[3.488]
∆Secondary School Enrollment	-0.040***	0.0002	-0.264***	-0.149***
	(-4.041) [0.00989]	(0.194) [0.000984]	(-23.35) [0.0113]	(-24.93) [0.00598]
Number of observations:	333	301	376	262
Number of groups:	50	41	54	38
Arellano – Bond test for	z = -1.94	z = -2.48	z = -2.07	z = -1.86
AR(1) in first differences:	Pr > z =0.0522	Pr > z = 0.0132	Pr > z =0.0389	Pr > z=0.0625
Arellano – Bond test for	z = -1.06	z = -1.67	z = -0.86	z = -1.11
AR(2) in first differences:	Pr > z =0.2897	Pr > z =0.0943	Pr > z=0.3913	Pr > z=0.2677
Sargan test:	Chi ² (54)	Chi ² (54)	Chi ² (54)	Chi ² (54)
	= 44.48	= 33.37	= 50.18	= 31.77

Table 1 (b): Coefficient of variation type financial openness and financial development indices

| Prob > chi ² |
|-------------------------|-------------------------|-------------------------|-------------------------|
| = 0.8187 | = 0.9878 | = 0.6226 | = 0.9932 |

Notes: All regressions are estimated using the Arellano – Bond dynamic panel GMM estimation with one lag of the dependent variable to be included in the model, 0 lags of the dependent variable and one lag of other variables to be used as instruments. The results reported here use the twostep estimator. The indices are constructed using the coefficient of variation methodology. The financial openness, banking sector development, bond and stock market development, financial development variables and institutional quality measures are constructed using coefficient of variation type indices. Figures in parentheses are t-statistics and the stars represent the significant t statistics for 1%, 5%, and 10% confidence levels respectively. The figures in brackets depict the standard errors for the coefficients.

	Dependent Variables				
	ΔBanking sector development	ΔBond market development	ΔStock market development	ΔFinancial development	
ΔDependent variable (Lagged)	0.854***	0.865***	0.367***	0.696***	
	(89.51)	(98.65)	(321.8)	(62.75)	
	[0.00954]	[0.00876]	[0.00114]	[0.0111]	
∆Financial openness	0.146***	0.0025	0.0691***	0.540***	
	(13.23)	(0.306)	(9.956)	(17.46)	
	[0.0110]	[0.00830]	[0.00694]	[0.0310]	
ATrada anonnass	0.0026***	0.0014***	0.0206***	0.0077***	
ΔTrade openness					
	(4.536)	(2.672)	(82.39)	(8.007)	
	[0.00581]	[0.000504]	[0.000250]	[0.000957] 0.587***	
ΔLog GDP per capita	0.0787	-0.368***	0.883***		
	(1.184)	(-12.32)	(16.70)	(12.36)	
	[0.0665]	[0.0298]	[0.0528]	[0.0475]	
ΔInstitutional Quality	-0.0696***	-0.0803***	-0.116***	-0.0004	
	(-4.401)	(-20.10)	(-8.887)	(-0.00870)	
	[0.0158]	[0.00400]	[0.0131]	[0.0502]	
ΔSecondary School Enrollment	0.0007**	0.0013***	-0.007***	-0.0019***	
	(2.223)	(5.595)	(-32.81)	(-3.102)	
	[0.000309]	[0.000229]	[0.000206]	[0.000596]	
Number of observations:	333	301	376	262	
Number of groups:	50	41	54	38	
Arellano – Bond test for	z = -2.22	z = -2.15	z = -2.06	z = -2.67	
AR(1) in first differences:	Pr > z=0.0266	Pr > z =0.0313	Pr > z =0.0396	Pr > z=0.0075	
Arellano – Bond test for	z = -1.36	z = -1.36	z = -0.91	z = -1.82	
AR(2) in first differences:	Pr > z=0.1733	Pr > z=0.1726	Pr > z=0.3652	Pr > z=0.0692	
Sargan test:	Chi ² (54)	Chi ² (54)	Chi ² (54)	Chi ² (54)	
	= 47.26	= 33.82	= 49.02	= 31.87	
	Prob > chi ²	Prob > chi ²	Prob > chi ²	Prob > chi ²	

Table 2 (a): First score financial openness and financial development indices

		= 0.7300	= 0.9857	= 0.6666	= 0.9929
--	--	----------	----------	----------	----------

Notes: All regressions are estimated using the Arellano – Bond dynamic panel GMM estimation with one lag of the dependent variable to be included in the model, 0 lags of the dependent variable and one lag of other variables to be used as instruments. The results reported here use the twostep estimator. The financial openness, banking sector development, bond and stock market development, financial development variables and institutional quality measures are constructed using first score of principal components. Figures in parentheses are t-statistics and the stars represent the significant t statistics for 1%, 5%, and 10% confidence levels respectively. The figures in brackets depict the standard errors for the coefficients.

		Depende	ent Variables	
	ΔBanking sector development	ΔBond market development	ΔStock market development	ΔFinancial development
ΔDependent variable (Lagged)	0.847***	0.808***	0.230***	0.302***
	(84.07)	(60.21)	(309.8)	(80.46)
	[0.0101]	[0.0134]	[0.000741]	[0.00376]
∆Financial openness	0.0983***	0.0970***	0.0825***	0.499***
	(12.86)	(16.51)	(6.954)	(37.54)
	[0.00765]	[0.00588]	[0.0119]	[0.0133]
∆Trade openness	0.0017***	0.00064***	0.0104***	0.0052***
	(4.218)	(6.969)	(55.70)	(9.015)
	[0.000397]	[0.000092]	[0.000187]	[0.000576]
ΔLog GDP per capita	0.131***	-0.147***	0.503***	0.477***
	(3.030)	(-6.496)	(19.32)	(9.976)
	[0.0431]	[0.00273]	[0.0260]	[0.0478]
∆Institutional Quality	-0.0088	-0.0551***	-0.106***	0.0359***
	-0.0088 (-1.378)	(-20.20)	(-13.01)	(5.299)
	[0.00638]	[0.00273]	[0.00812]	[0.00678]
ΔSecondary School Enrollment	0.0008***		-0.0045***	0.0021***
	(4.181)	0.00001 (0.274)	(-23.17)	(5.550)
	[0.000190]	[0.000120]	[0.000196]	[0.000386]
Number of observations:	333	301	376	262
Number of groups:	50	41	54	38
Arellano – Bond test for	z = -2.47	z = -2.39	z = -1.90	z = -1.99
AR(1) in first differences:	Pr > z =0.0134	Pr > z =0.0168	Pr > z =0.0573	Pr >z=0.0468
Arellano – Bond test for	z = -1.46	z = -1.66	z = -0.96	z = -1.07
AR(2) in first differences:	Pr > z =0.1435	Pr > z=0.0974	Pr > z=0.3386	Pr >z=0.2854
Sargan test:	Chi ² (54)	Chi ² (54)	Chi ² (54)	Chi ² (54)
	= 42.13	= 37.89	= 44.36	= 33.49
	Prob > chi ²	Prob > chi ²	Prob > chi ²	Prob > chi ²

Table 2 (b): Principal components (information from all components) type financial openness and financial development indices

	= 0).8795 =	0.9528 =	= 0.8222	= 0.9872
--	-----	----------	----------	----------	----------

Notes: All regressions are estimated using the Arellano – Bond dynamic panel GMM estimation with one lag of the dependent variable to be included in the model, 0 lags of the dependent variable and one lag of other variables to be used as instruments. The results reported here use the twostep estimator. The indices are constructed following the principal component methodology that utilizes all components. Figures in parentheses are t-statistics and the stars represent the significant t statistics for 1%, 5%, and 10% confidence levels respectively. The figures in brackets depict the standard errors for the coefficients.

		Dependent Variables				
	ΔBanking sector development	ΔBond market development	ΔStock market development	ΔFinancial development		
ΔDependent variable (Lagged)	0.848***	0.808***	0.239***	0.312***		
	(66.15)	(59.66)	(161.4)	(31.20)		
	[0.0128]	[0.0135]	[0.00148]	[0.0100]		
ΔFinancial openness	0.0752***	0.170***	0.384***	0.480***		
	(2.722)	(16.39)	(19.58)	(17.70)		
	[0.0276]	[0.0104]	[0.0196]	[0.0271]		
ATrada anonnoss	0.0015***	0.0011***	0.0102***	0.0055***		
ΔTrade openness		(6.581)	(70.19)			
	(3.667)			(9.982)		
Alog CDD por capita	[0.000416] 0.149***	[0.000163] -0.131***	[0.000146] 0.448***	[0.000547] 0.486***		
ΔLog GDP per capita						
	(2.968)	(-4.972)	(13.12)	(7.235)		
	[0.0502]	[0.0263]	[0.0341]	[0.0672]		
ΔInstitutional Quality	-0.0172	-0.0439***	-0.0387**	0.0505***		
	(-1.323)	(-11.81)	(-5.265)	(4.214)		
	[0.0130]	[0.00372]	[0.00735]	[0.0120]		
ΔSecondary School Enrollment	0.0009***	0.000135	-0.0031***	0.00234***		
	(5.709)	(0.818)	(-9.851)	(7.839)		
	[0.000162]	[0.000165]	[0.000316]	[0.000299]		
ΔInteraction term	0.00035*	-0.00086***	-0.0012***	0.00002		
	(1.907)	(-11.98)	(-23.99)	(0.0763)		
	[0.000182]	[0.0000716]	[0.0000487]	[0.000282]		
Number of observations:	333	301	376	262		
Number of groups:	50	41	54	38		
Arellano – Bond test for	z = -2.50	z = -2.38	z = -1.91	z = -2.03		
AR(1) in first differences:	Pr > z =0.0126	Pr > z = 0.0175	Pr > z =0.0567	Pr > z=0.0424		
Arellano – Bond test for	z = -1.47	z = -1.73	z = -1.00	z = -1.04		
AR(2) in first differences:	Pr > z =0.1415	Pr > z =0.0844	Pr > z=0.3163	Pr > z=0.2990		

Table 3: Principal components type financial openness and financial development indices with the interaction term

Sargan test:	Chi ² (54)	Chi ² (54)	Chi ² (54)	Chi ² (54)
	= 47.62	= 36.44	= 47.84	= 35.02
	Prob > chi ²			
	= 0.7173	= 0.9680	= 0.7096	= 0.9790

Notes: All regressions are estimated using the Arellano – Bond dynamic panel GMM estimation with one lag of the dependent variable to be included in the model, 0 lags of the dependent variable and one lag of other variables to be used as instruments. The results reported here use the twostep estimator. The indices are constructed following the principal component methodology that utilizes all components. The financial openness, banking sector development, bond and stock market development, financial development variables and institutional quality measures are constructed using principal component indices. Figures in parentheses are t-statistics and the stars represent the significant t statistics for 1%, 5%, and 10% confidence levels respectively. The figures in brackets depict the standard errors for the coefficients.

		Dependent Variables				
	ΔBanking sector development	ΔBond market development	ΔStock market development	ΔFinancial development		
∆Dependent variable (Lag 1)	1.1399***	0.4776***	0.2363***	0.3271***		
(8)	(46.01)	(20.75)	(122.07)	(41.72)		
	[0.0247731]	[0.0230098]	[0.0019359]	[0.0078403]		
∆Dependent variable (Lag 2)	-0.3948***	-0.0416***	-0.1279***	-0.0764***		
((-20.27)	(-7.32)	(-49.68)	(-6.38)		
	[0.019475]	[0.0056819]	[0.0025747]	[0.0119731]		
∆Dependent variable (Lag 3)	0.1837***	0.3101***	-0.0491***	-0.1181***		
((() (()(((19.97)	(57.05)	(-21.95)	(-10.58)		
	[0.0092001]	[0.0054356]	[0.0022359]	[0.0111681]		
∆Financial openness	0.0947***	0.1237***	0.0689***	0.476***		
	(8.03)	(13.86)	(4.08)	(12.13)		
	[0.01177981]	[0.0089279]	[0.168978]	[0.0392724]		
∆Trade openness	0.0004	-0.0003**	0.0131***	0.0057***		
	(0.68)	(-2.42)	(67.50)	(6.82)		
	[0.000539]	[0.0001152]	[0.0001944]	[0.0008389]		
∆Log GDP per capita	0.0569*	-0.0807***	-0.1367**	0.617***		
	(1.90)	(-2.90)	(-2.22)	(3.62)		
	[0.0299644]	[0.0278365]	[0.0616486]	[0.1706641]		
∆Institutional Quality	-0.01106	-0.04822***	-0.2413***	-0.0129		
Anstitutional Quarty	(-0.45)	(-18.95)	(-8.50)	(-0.50)		
	[0.0246444]	[0.0025442]	[0.028378]	[0.0257977]		
∆Secondary School Enrollment	0.0004	-0.0016***	-0.0027***	0.0005		
	(1.48)	(-11.62)	(-5.13)	(1.05)		
	[0.0002368]	[0.0001344]	[0.0004416]	[0.0004206]		
Number of observations:	289	274	346	225		
Number of groups:	49	40	53	37		
Arellano – Bond test for	z = -3.04	z = -1.40	z = -1.96	z = -1.96		
	Pr > z =0.0024	Pr > z = 0.1628	Pr > z=0.0501	Pr >z=0.0498		

Table 4: Principal components type financial openness and financial development indices; Changing lag structure

AR(1) in first differences:	z = 0.67	z = 1.03	z = 1.38	z = 1.11
Arellano – Bond test for	Pr > z =0.5006	Pr > z =0.3012	Pr >z=0.1689	Pr >z=0.2668
AR(2) in first differences:				
Sargan test:	Chi ² (49)	Chi ² (49)	Chi ² (49)	Chi ² (49)
	= 44.27	= 32.75	= 47.93	= 31.29
	Prob > chi ²			
	= 0.6650	= 0.9640	= 0.5165	= 0.9771

Notes: All regressions are estimated using the Arellano – Bond dynamic panel GMM estimation with 3 lags of the dependent variable to be included in the model, 0 lags of the dependent variable and 3 lags of other variables to be used as instruments. The results reported here use the twostep estimator. The indices are constructed following the principal component methodology that utilizes all components. The indices are constructed following the principal component methodology that utilizes all components. The indices are constructed following the principal component methodology that utilizes are constructed using principal component indices. Figures in parentheses are t-statistics and the stars represent the significant t statistics for 1%, 5%, and 10% confidence levels respectively. The figures in brackets depict the standard errors for the coefficients.

		Depende	ent Variables		
	ΔBanking sector development	-		∆Financial development	
ΔDependent variable (Lagged)	0.809***	0.5387***	0.2234***	0.414***	
	(31.16)	(14.27)	(31.09)	(5.07)	
	[0.0259655]	[0.0377382]	[0.0071859]	[0.0816462]	
∆Financial openness	0.0334**	0.0055	0.0941***	0.1138***	
	(2.09)	(0.56)	(6.28)	(4.94)	
	[0.0160238]	[0.0098091]	[0.0149839]	[0.0230477]	
∆Trade openness	-0.0018**	-0.0013***	0.0118***	0.0029*	
	(-2.55)	(-3.16)	(16.48)	(1.70)	
	[0.0006937]	[0.0004178]	[0.0007136]	[0.0016757]	
Allog CDB por conito	0.5396***	0.0541	0.3992**	0.7263***	
ΔLog GDP per capita	(3.98)	(0.57)	(2.42)	(5.81)	
	[0.1357218]	[0.0943717]	[0.1648696]	[0.1250271]	
Alastitutional Quality	0.0038	-0.2643***	-0.072***	-0.1168***	
ΔInstitutional Quality					
	(0.32)	(-18.88)	(-3.11)	(-3.32)	
	[0.0119641]	[0.0140001]	[0.0231092]	[0.0351751]	
ΔSecondary School Enrollment	-0.0029	0.0048***	-0.006***	-0.0015	
	(-1.58)	(6.39)	(-5.44)	(-0.76)	
	[0.0018469]	[0.0007486]	[0.0011049]	[0.0020077]	
Number of observations:	153	112	163	100	
Number of groups:	25	18	27	16	
Arellano – Bond test for	z = -2.04	z = -1.90	z = -1.88	z = -1.80	
AR(1) in first differences:	Pr > z =0.0416	Pr > z = 0.0571	Pr > z =0.0603	Pr > z=0.0726	
Arellano – Bond test for	z = 1.32	z = 1.33	z = -1.31	z = -0.08	
AR(2) in first differences:	Pr > z =0.1854	Pr > z =0.1844	Pr > z=0.1907	Pr > z=0.9342	
Sargan test:	Chi ² (54)	Chi ² (54)	Chi ² (54)	Chi ² (54)	
	= 21.00	= 14.02	= 18.40	= 8.21	
	Prob > chi ²	Prob > chi ²	Prob > chi ²	Prob > chi ²	

Table 5: Principal components type financial openness and financial development indices for the developing country sample

= 1.0000 = 1.0000 = 1.0000 = 1.0000

Notes: All regressions are estimated using the Arellano – Bond dynamic panel GMM estimation with one lag of the dependent variable to be included in the model, 0 lags of the dependent variable and one lag of other variables to be used as instruments. The results reported here use the twostep estimator. The indices are constructed following the principal component methodology that utilizes all components. The indices are constructed following the principal component methodology that utilizes all components. Figures in parentheses are t-statistics and the stars represent the significant t statistics for 1%, 5%, and 10% confidence levels respectively. The figures in brackets depict the standard errors for the coefficients.

		Depende	nt Variables		
	ΔBanking sector development	-		ΔFinancial development	
ΔDependent variable (Lagged)	0.690***	0.908***	0.225***	0.388***	
	(140.4)	(94.97)	(235.8)	(59.98)	
	[0.00492]	[0.00956]	[0.000954]	[0.00647]	
ΔΚΑΟΡΕΝ	0.0965***	-0.0304***	0.0545***	-0.0273***	
	(18.75)	(-13.22)	(5.969)	(-6.925)	
	[0.00515]	[0.00230]	[0.00913]	[0.00395]	
ΔTrade openness	0.0038***	0.0016***	0.0123***	0.0103***	
·	(20.82)	(13.62)	(74.08)	(23.23)	
	[0.000180]	[0.000116]	[0.000166]	[0.000442]	
ΔLog GDP per capita	0.181***	-0.218***	0.602***	0.674***	
0	(4.442)	(-10.62)	(29.70)	(15.10)	
	[0.0408]	[0.0205]	[0.0203]	[0.0446]	
ΔInstitutional Quality	-0.0873***	-0.0613***	-0.140***	-0.0425***	
-	(-24.19)	(-20.42)	(-19.23)	(-3.385)	
	[0.00361]	[0.00300]	[0.00726]	[0.0125]	
ΔSecondary School Enrollment	0.0001	-0.0002***	-0.0055***	-0.0004**	
-	(0.734)	(-3.004)	(-28.48)	(-2.447)	
	[0.000185]	[7.04e-05]	[0.000193]	[0.000143]	
Number of observations:	373	318	420	276	
Number of groups:	54	41	57	38	
Arellano – Bond test for	z = -1.06	z = -2.50	z = -1.93	z = -2.07	
AR(1) in first differences:	Pr > z = 0.2881	Pr > z = 0.0124	Pr > z = 0.0541	Pr > z = 0.0387	
Arellano – Bond test for	z = 0.87	z = -2.03	z = -1.01	z = -0.45	
AR(2) in first differences:	Pr > z = 0.3831	Pr > z = 0.0428	Pr > z = 0.3112	Pr > z = 0.6558	
Sargan test:	Chi ² (54)= 51.66	Chi ² (54)= 36.78	Chi ² (54)= 51.03	Chi ² (54)= 35.27	
	Prob > chi ²	Prob > chi ²	Prob > chi ²	Prob > chi ²	

 Table 6: Principal components type financial development indices and the Chinn and Ito index of capital account openness

= 0.5651	= 0.9648	= 0.5897	= 0.9773	

Notes: All regressions are estimated using the Arellano – Bond dynamic panel GMM estimation with one lag of the dependent variable to be included in the model, 0 lags of the dependent variable and one lag of other variables to be used as instruments. The results reported here use the twostep estimator. The indices for banking sector, bond and stock market and overall financial development are constructed following the principal component methodology that utilizes all components. KAOPEN is the Chinn and Ito index of capital account openness. The indices are constructed following the principal component methodology that utilizes all components. Figures in parentheses are t-statistics and the stars represent the significant t statistics for 1%, 5%, and 10% confidence levels respectively. The figures in brackets depict the standard errors for the coefficients.

	Dependent Variables							
	ΔLiquid Liabilities	ΔPrivate credit	ΔDeposit bank assets	ΔTotal bank assets	ΔDomestic credit	ΔStock market capitalization	ΔTurnover ratio	ΔValue traded
ΔDependent variable (Lagged)	0.601***	0.609***	0.150***	0.561***	0.982***	0.500***	0.0985***	0.485***
	(52.84)	(277.7)	(173.2)	(100.4)	(243.7)	(199.9)	(137.5)	(3,819)
	[0.0114]	[0.00219]	[0.000865]	[0.00559]	[0.00403]	[0.00250]	[0.000716]	[0.000127]
ΔΚΑΟΡΕΝ	1.013***	1.362***	-1.113***	1.053***	0.676***	-8.854***	13.08***	-6.110***
	(6.273)	(13.17)	(-68.14)	(6.970)	(3.418)	(-29.47)	(37.62)	(-27.17)
	[0.162]	[0.103]	[0.0163]	[0.151]	[0.198]	[0.300]	[0.348]	[0.225]
ΔTrade openness	0.0656***	0.105***	-0.0283***	0.187***	0.0534***	0.553***	0.499***	1.096***
	(10.17)	(20.29)	(-84.20)	(39.09)	(9.375)	(35.09)	(57.29)	(1,256)
	[0.00645]	[0.00517]	[0.000336]	[0.00479]	[0.00569]	[0.0158]	[0.00870]	[0.000873]
ΔLog GDP per capita	6.465***	7.489***	17.83***	3.550*	-4.777***	64.59***	-50.81***	114.5***
	(9.703)	(8.143)	(208.3)	(1.942)	(-2.748)	(40.68)	(-61.01)	(512.4)
	[0.666]	[0.920]	[0.0856]	[1.828]	[1.738]	[1.588]	[0.833]	[0.223]
∆Institutional Quality	-2.448***	-2.168***	0.0457**	-0.414***	-4.953***	-10.84***	-14.48***	-6.836***
	(-16.67)	(-22.12)	(2.406)	(-3.957)	(-30.60)	(-31.90)	(-92.57)	(-42.26)
	[0.147]	[0.0980]	[0.0190]	[0.105]	[0.162]	[0.340]	[0.156]	[0.162]
∆Secondary School Enrollment	-0.0517***	-0.0759***	0.0801***	0.0004	-0.0223***	-0.0150**	-0.442***	-0.458***
	(-13.58)	(-20.59)	(35.14)	(0.100)	(-6.438)	(-2.080)	(-99.89)	(-129.5)
	[0.00381]	[0.00369]	[0.00228]	[0.00426]	[0.00346]	[0.00722]	[0.00442]	[0.00354]
Number of observations:	399	408	386	376	411	420	420	420
Number of groups:	56	56	55	54	57	57	57	57
Arellano – Bond test for	z = -0.78	z = 0.31	z = -0.97	z = 0.93	z = -2.12	z = 0.18	z = -1.54	z = -2.02
AR(1) in first differences:	Pr > z =0.4337	Pr > z = 0.7551	Pr > z = 0.3325	Pr > z = 0.3531	Pr > z = 0.0339	Pr > z = 0.8564	Pr > z = 0.1224	Pr > z = 0.0439
Arellano – Bond test for	z = -1.85	z = -1.79	z = 1.07	z = -1.66	z = 1.23	z = -1.96	z = -1.07	z = 0.18
AR(2) in first differences:	Pr > z =0.0641	Pr > z = 0.0732	Pr > z = 0.2883	Pr > z = 0.0961	Pr > z = 0.2174	Pr > z = 0.0495	Pr > z = 0.2865	Pr > z = 0.8570
Sargan test:	Chi ² (54)=53.57	Chi ² (54)= 52.70	Chi ² (54)= 48.90	Chi ² (54)= 49.00	Chi ² (54)= 53.78	Chi ² (54)= 52.34	Chi ² (54)= 52.55	Chi ² (54)= 53.9
	Prob > chi ²	Prob > chi ²	Prob > chi ²	Prob > chi ²	Prob > chi ²	Prob > chi ²	Prob > chi ²	Prob > chi ²

Table 7: Individual financial development measures and the Chinn and Ito index of capital account openness

= 0.4909 = 0.5247 = 0.6708 = 0.6673 = 0.4828 = 0.5386 = 0.5306 = 0.477	/0
--	----

Notes: All regressions are estimated using the Arellano – Bond dynamic panel GMM estimation with one lag of the dependent variable to be included in the model, 0 lags of the dependent variable and one lag of other variables to be used as instruments. The results reported here use the twostep estimator. The indices for banking sector, bond and stock market and overall financial development are constructed following the principal component methodology that utilizes all components. KAOPEN is the Chinn and Ito index of capital account openness, the dependent variables are: Liquid liabilities (% of GDP), Private credit by deposit money banks and other institutions (% of GDP), Deposit money bank assets to the sum of deposit money bank assets and central bank assets (in percentages), Total bank assets (% of GDP). Domestic credit provided by the banking sector (% of GDP), Stock market capitalization (% of GDP), Stock market turnover ratio (in percentages), Stock market value traded (% of GDP). Figures in parentheses are t-statistics and the stars represent the significant t statistics for 1%, 5%, and 10% confidence levels respectively. The figures in brackets depict the standard errors for the coefficients.

	Dependent Variables							
	ΔLiquid liabilities	ΔPrivate credit	∆Deposit bank assets	ΔTotal bank assets	ΔDomestic credit	ΔStock market capitalization	ΔTurnover ratio	ΔValue traded
ΔDependent variable (Lagged)	0.8998***	0.6198***	0.7784***	0.572***	0.9748***	0.522***	0.0998***	0.5103***
	(454.64)	(143.51)	(481.82)	(149.82)	(102.77)	(173.98)	(361.07)	(1809.72)
	[0.0019791]	[0.004319]	[0.0016155]	[0.0038166]	[0.0094847]	[0.0029979]	[0.0002764]	[0.000282]
ΔFinancial openness	0.7643***	3.7167***	-1.158***	9.509***	0.0235	2.934***	4.9186***	6.661***
	(55.85)	(45.67)	(-76.08)	(53.13)	(0.21)	(27.67)	(32.15)	(52.82)
	[0.0136861]	[0.0813903]	[0.0152147]	[0.1789926]	[0.1123886]	[0.1060275]	[0.1530111]	[0.1261231]
∆Trade openness	0.0730***	0.1259***	-0.0148***	0.1391***	0.1283**	0.7294***	0.3289***	1.0343***
	(9.28)	(14.75)	(-24.75)	(17.76)	(21.17)	(71.12)	(19.50)	(350.27)
	[0.0078662]	[0.0085404]	[0.0005986]	[0.0077456]	[0.0060617]	[0.0102561]	[0.0168636]	[0.0029528]
ΔLog GDP per capitat	3.1754***	9.476***	5.831***	0.212	-1.81506	36.911***	-53.90***	95.845***
	(2.86)	(6.58)	(65.41)	(0.14)	(-1.26)	(30.61)	(-55.64)	(129.62)
	[1.111859]	[1.440603]	[0.0891392]	[1.530057]	[1.440914]	[1.205813]	[0.9687153]	[0.7394168]
∆Institutional Quality	-4.687***	-2.935***	1.0541***	2.631***	-7.934***	-15.222***	-12.961***	-3.973***
Emstitutional Quality	(-16.31)	(-24.27)	(49.91)	(16.04)	(-25.33)	(-59.65)	(-46.04)	(-12.57)
	[0.2873217]	[0.1208917]	[0.021122]	[0.16396]	[0.313209]	[0.2551843]	[0.28153]	[0.3161314]
∆Secondary School Enrollment	-0.0948**	-0.0628***	0.00366***	0.0695***	-0.092***	-0.01422*	-0.427***	-0.392***
	(-8.17)	(-15.35)	(17.00)	(36.40)	(-21.28)	(-1.88)	(-27.43)	(-119.80)
	[0.0115953]	[0.0040934]	[0.0002152]	[0.0019105]	[0.0042989]	[0.0075512]	[0.0155645]	[0.0032682]
	[0.0113933]	[0.0040934]	[0.0002132]	[0.0019103]	[0.0042989]	[0.0075512]	[0.0133043]	[0.0052082]
Number of observations:	357	365	344	335	368	376	376	376
Number of groups:	53	53	51	50	54	54	54	54
Arellano – Bond test for	z = -0.50	z = -0.16	z = -2.88	z = 0.52	z = -2.06	z = 0.54	z = -1.53	z = -1.93
AR(1) in first differences:	Pr > z=0.6200	Pr > z=0.8734	Pr >z=0.0039	Pr > z=0.5997	Pr >z =0.0390	Pr >z=0.5878	Pr>z=0.1256	Pr >z=0.053
Arellano – Bond test for	z = -1.63	z = -1.55	z = -1.63	z = -1.29	z = 1.67	z = -2.20	z = -1.06	z = 0.25
AR(2) in first differences:	Pr>z=0.1024	Pr > z=0.1208	Pr >z=0.1027	Pr > z=0.1982	Pr >z =0.0942	Pr >z=0.0275	Pr>z=0.2914	Pr >z=0.8056
Sargan test:	Chi ² (54)	Chi ² (54)	Chi ² (54)	Chi ² (54)	Chi ² (54)	Chi ² (54)	Chi ² (54)	Chi ² (54)
	= 50.84	= 46.85	= 44.53	= 47.30	= 51.00	= 48.15	= 45.83	= 50.40
	Prob >chi ²	Prob >chi ²	Prob >chi ²	Prob >chi ²	Prob > chi ²		Prob > chi ²	Prob > chi ²

Table 8: Principal component type financial openness index and individual financial development measures

= 0.5970	= 0.7441	= 0.8175	= 0.7286	= 0.5908	Prob > chi ²	= 0.7776	= 0.6139
					= 0.6984		

Notes: All regressions are estimated using the Arellano – Bond dynamic panel GMM estimation with one lag of the dependent variable to be included in the model, 0 lags of the dependent variable and one lag of other variables to be used as instruments. The results reported here use the twostep estimator. The indices are constructed following the principal component methodology that utilizes all components. The variables are: Liquid liabilities (% of GDP), Private credit by deposit money banks and other institutions (% of GDP), The ratio of deposit money bank assets to the sum of deposit money bank assets to central bank assets (in percentages), Total bank assets (% of GDP), Domestic credit provided by the banking sector (% of GDP), Stock market capitalization (% of GDP), Stock market turnover ratio (in percentages), and Stock market total value traded (% of GDP). Figures in parentheses are t-statistics and the stars represent the significant t statistics for 1%, 5%, and 10% confidence levels respectively. The figures in brackets depict the standard errors for the coefficients.